Author’s response to reviews

Title: Improving quality of life through the routine use of the Patient Concerns Inventory for head and neck cancer patients: a cluster preference randomized controlled trial

Authors:

Simon Rogers (snrogers.aintree@gmail.com)
Derek Lowe (astraglobeltd@btconnect.com)
Chermaine Lowies (chermaine.lowies@aintree.nhs.uk)
Seow Yeo (s.t.yeo@bangor.ac.uk)
Christine Allmark (allmarkc@yahoo.co.uk)
Dominic Macareavy (dominic.macareavy@btinternet.com)
Gerald Humphris (gmh4@st-andrews.ac.uk)
Robert Flavel (robertflavel@btinternet.com)
Cherith Semple (Cherith.Semple@setrust.hscni.net)
Stephen Thomas (steve.thomas@bristol.ac.uk)
Anastasios Kanatas (a.kanatas@doctors.org.uk)

Version: 1  Date: 17 Sep 2017

Author’s response to reviews:

In response to the email 15th September 2017-09-14Fwd: Your submission to BMC Cancer - BCAN-D-17-01469

1. Ethical and Funding Approval Documentation

Copies of all ethical approval and funding approval have been emailed to you by Miss C Lowies (chermaine.lowies@aintree.nhs.uk) to the following email address: BMCSeriesEditorial@biomedcentral.com.
2. Funding

The study has been externally funded by a major funding body, NIHR, Research for Patient Benefit (RfPB), trial registration: 32382. The study protocol has undergone peer-review by the funding body. The study has not received funding/assistance from a commercial organization.

3. Study status

The study is ongoing with recruitment started 4 months ago.

4. Related Articles

No other publications containing the results of this study have already been published or submitted to any journal.

Yours faithfully

Simon N Rogers